Summary

ffl charged particle pseudorapidity density

– width of the distribution decreases

from 1.7 to 1.45 with centrality

ffl spectra of h^�, π^+ and π^�

– continuous change of slope with m_T – smaller change for central collisions

ffl spectra of (+) � (�)

– change of slope with y reflecting change with m_T – nearly no change of slope with centrality

for mT � m0 > 0.7GeV/c

– preliminary rapidity distribution

ffl anisotropic flow

– directed flow (max 3.5%) – in plane elliptic flow ( max 4.5%) – centrality and pseudorapidity dependence